Author's response to reviews

Title: Different skeletal effects of the peroxisome proliferator activated receptor (PPAR)alpha agonist fenofibrate and the PPAR gamma agonist pioglitazone

Authors:

Unni Syversen (unni.syversen@ntnu.no)
Astrid K Stunes (kamilla.stunes@ntnu.no)
Bjørn I Gustafsson (bjorn.gustafsson@ntnu.no)
Karl J Obrant (karl.obrant@orto.mas.lu.se)
Lars Nordsletten (lars.nordsletten@medisin.uio.no)
Berge Rolf (rolf.berge@med.uib.no)
Liv Thommesen (liv.thommesen@ntnu.no)
Janne E Reseland (j.e.reseland@odont.uio.no)

Version: 4 Date: 27 March 2009

Author's response to reviews: see over
Dear Editor

Enclosed, please find our revised manuscript entitled: “Different skeletal effects of the peroxisome proliferator activated receptor (PPAR) alpha agonist fenofibrate and the PPAR gamma agonist pioglitazone” by Unni Syversen, Astrid K Stunes, Björn I Gustafsson, Karl J Obrant, Lars Nordsletten, Rolf Berge, Liv Thommesen and Janne E Reseland.

Major and minor revisions have been made according to the requested instructions as follows:

Author list - The list of authors in the manuscript should be written exactly as they are in the submission system. The authors (forename, initial (if any), surname) should be written identically and in the order in which they are listed in the submission system. Done

Conclusions - Please insert a Conclusions section after the Discussion. We recommend including a Conclusions heading above the paragraph which beings "In conclusion..." Done

Reference 26 & please provide full details for this reference. This means for an Article providing volume/issue/page numbers and for a book the publisher/city. Done

Reference 40 & please provide full details for this reference. This means for an Article providing volume/issue/page numbers and for a book the publisher/city. Done

Tables - Please do not include vertical lines in your tables. Done, table 5 is rewritten.

Figure size - at 100% the text is not readable, we recommend you upload a higher resolution version of the figure. New figures are uploaded with higher resolution.

Figure title - The legend and title should be part of the manuscript file after the reference list. Please provide a legend for each figure uploaded. Figure legends for figure 1, 2 and 3 are included after references in the main manuscript.
Minor revisions (We can make these changes on resubmission, although it will speed up publication of your manuscript if you do them while making the major changes above).

Title – Please use sentence case in the title, both in the manuscript file and the submission system i.e. remove all unnecessary capitals (capitalise only the first word, and proper nouns, and no full stops/periods). For example you would change `The Title of My Prestigious Publication: A Scientific Study¿. To `The title of my prestigious publication: a scientific study¿. Done

Keywords, and Running title ¿ Please remove these. Done

Introduction – Please rename this section `Background¿. Done

Materials & Methods – Please rename this section 'Methods'. Done

Authors¿ contributions – please ensure this is formatted into a paragraph rather than a list or in bullet points. This section is rewritten

Table title and legend – The table title should appear above the table and should be only one sentence long. Please place any other text as a legend (or key), which should appear below the table. Done, all tables are changed according to this.

Thank you for publishing our manuscript!

On behalf of the authors

Astrid Kamilla Stunes